\W\begin{center}
\W\begin{table}[!ht]
\W\xlink{mk23a.txt}{\rawdatadir/mk23a.txt}~Guirao (1982)~\cite{c_guir82}
\W\label{mk23a}
\W\end{table}
\W\end{center}
\T\begin{longtable}{ccc}
\T\caption{mk23a~\protect\cite[Guirao (1982)]{c_guir82}\label{mk23a}}\\
\T\input{\datadir/mk23a.txt}
\T\end{longtable}

\begin{center}
\begin{tabular}{rlrl}
Category: & critical tube & Fuel: & H2 \\
Sub-Category: &  & Oxidizer: & Air \\
Initial Pressure: & 101.3 kPa & Diluent: &  \\
Initial Temperature: & 293 K & Equivalence Ratio: & 0.85-1.86 \\
\end{tabular}
\end{center}

\W\begin{center}
\W\begin{table}[!ht]
\W\xlink{mk23b.txt}{\rawdatadir/mk23b.txt}~Guirao (1982)~\cite{c_guir82}
\W\label{mk23b}
\W\end{table}
\W\end{center}
\T\begin{longtable}{ccc}
\T\caption{mk23b~\protect\cite[Guirao (1982)]{c_guir82}\label{mk23b}}\\
\T\input{\datadir/mk23b.txt}
\T\end{longtable}

\begin{center}
\begin{tabular}{rlrl}
Category: & critical tube & Fuel: & H2 \\
Sub-Category: &  & Oxidizer: & Air \\
Initial Pressure: & 101.3 kPa & Diluent: &  \\
Initial Temperature: & 293 K & Equivalence Ratio: & 0.85-1.86 \\
\end{tabular}
\end{center}

\W\begin{center}
\W\begin{table}[!ht]
\W\xlink{at72a.txt}{\rawdatadir/at72a.txt}~Knystautas (1982)~\cite{r_knys82}
\W\label{at72a}
\W\end{table}
\W\end{center}
\T\begin{longtable}{ccc}
\T\caption{at72a~\protect\cite[Knystautas (1982)]{r_knys82}\label{at72a}}\\
\T\input{\datadir/at72a.txt}
\T\end{longtable}

\begin{center}
\begin{tabular}{rlrl}
Category: & critical tube & Fuel: & H2 \\
Sub-Category: &  & Oxidizer: & O2 \\
Initial Pressure: & 101.3 kPa & Diluent: & N2 \\
Initial Temperature: & 293 K & Equivalence Ratio: & 1 \\
\end{tabular}
\end{center}

\W\begin{center}
\W\begin{table}[!ht]
\W\xlink{at56a.txt}{\rawdatadir/at56a.txt}~Makris (1994)~\cite{a_makr94}
\W\label{at56a}
\W\end{table}
\W\end{center}
\T\begin{longtable}{cc}
\T\caption{at56a~\protect\cite[Makris (1994)]{a_makr94}\label{at56a}}\\
\T\input{\datadir/at56a.txt}
\T\end{longtable}

\begin{center}
\begin{tabular}{rlrl}
Category: & critical tube & Fuel: & H2 \\
Sub-Category: &  & Oxidizer: & O2 \\
Initial Pressure: & 101.3 kPa & Diluent: &  \\
Initial Temperature: & 293 K & Equivalence Ratio: & .6-1.2 \\
\end{tabular}
\end{center}

\W\begin{center}
\W\begin{table}[!ht]
\W\xlink{at75a.txt}{\rawdatadir/at75a.txt}~Matsui (1979)~\cite{h_mats79}
\W\label{at75a}
\W\end{table}
\W\end{center}
\T\begin{longtable}{ccc}
\T\caption{at75a~\protect\cite[Matsui (1979)]{h_mats79}\label{at75a}}\\
\T\input{\datadir/at75a.txt}
\T\end{longtable}

\begin{center}
\begin{tabular}{rlrl}
Category: & critical tube & Fuel: & H2 \\
Sub-Category: &  & Oxidizer: & O2 \\
Initial Pressure: & 101.3 kPa & Diluent: &  \\
Initial Temperature: & 293 K & Equivalence Ratio: & 1 \\
\end{tabular}
\end{center}

\W\begin{center}
\W\begin{table}[!ht]
\W\xlink{at72c.txt}{\rawdatadir/at72c.txt}~Matsui (1979)~\cite{h_mats79}
\W\label{at72c}
\W\end{table}
\W\end{center}
\T\begin{longtable}{ccc}
\T\caption{at72c~\protect\cite[Matsui (1979)]{h_mats79}\label{at72c}}\\
\T\input{\datadir/at72c.txt}
\T\end{longtable}

\begin{center}
\begin{tabular}{rlrl}
Category: & critical tube & Fuel: & H2 \\
Sub-Category: &  & Oxidizer: & O2 \\
Initial Pressure: & 101.3 kPa & Diluent: & N2 \\
Initial Temperature: & 293 K & Equivalence Ratio: & 1 \\
\end{tabular}
\end{center}

\W\begin{center}
\W\begin{table}[!ht]
\W\xlink{at75b.txt}{\rawdatadir/at75b.txt}~Moen (1985)~\cite{i_moen85}
\W\label{at75b}
\W\end{table}
\W\end{center}
\T\begin{longtable}{ccc}
\T\caption{at75b~\protect\cite[Moen (1985)]{i_moen85}\label{at75b}}\\
\T\input{\datadir/at75b.txt}
\T\end{longtable}

\begin{center}
\begin{tabular}{rlrl}
Category: & critical tube & Fuel: & H2 \\
Sub-Category: &  & Oxidizer: & O2 \\
Initial Pressure: & 40-80 kPa & Diluent: &  \\
Initial Temperature: & 293 K & Equivalence Ratio: & 1 \\
\end{tabular}
\end{center}

\W\begin{center}
\W\begin{table}[!ht]
\W\xlink{at75d.txt}{\rawdatadir/at75d.txt}~Moen (1985)~\cite{i_moen85}
\W\label{at75d}
\W\end{table}
\W\end{center}
\T\begin{longtable}{ccc}
\T\caption{at75d~\protect\cite[Moen (1985)]{i_moen85}\label{at75d}}\\
\T\input{\datadir/at75d.txt}
\T\end{longtable}

\begin{center}
\begin{tabular}{rlrl}
Category: & critical tube & Fuel: & H2 \\
Sub-Category: &  & Oxidizer: & O2 \\
Initial Pressure: & 40-80 kPa & Diluent: & CF3Br \\
Initial Temperature: & 293 K & Equivalence Ratio: & 1 \\
\end{tabular}
\end{center}

\W\begin{center}
\W\begin{table}[!ht]
\W\xlink{at75c.txt}{\rawdatadir/at75c.txt}~Moen (1985)~\cite{i_moen85}
\W\label{at75c}
\W\end{table}
\W\end{center}
\T\begin{longtable}{ccc}
\T\caption{at75c~\protect\cite[Moen (1985)]{i_moen85}\label{at75c}}\\
\T\input{\datadir/at75c.txt}
\T\end{longtable}

\begin{center}
\begin{tabular}{rlrl}
Category: & critical tube & Fuel: & H2 \\
Sub-Category: &  & Oxidizer: & O2 \\
Initial Pressure: & 80 kPa & Diluent: & CO2 \\
Initial Temperature: & 293 K & Equivalence Ratio: & 1 \\
\end{tabular}
\end{center}

\W\begin{center}
\W\begin{table}[!ht]
\W\xlink{at43.txt}{\rawdatadir/at43.txt}~Plaster (1991)~\cite{m_plas91}
\W\label{at43}
\W\end{table}
\W\end{center}
\T\begin{longtable}{cc}
\T\caption{at43~\protect\cite[Plaster (1991)]{m_plas91}\label{at43}}\\
\T\input{\datadir/at43.txt}
\T\end{longtable}

\begin{center}
\begin{tabular}{rlrl}
Category: & critical tube & Fuel: & H2 \\
Sub-Category: &  & Oxidizer: & O2 \\
Initial Pressure: & 100 kPa & Diluent: &  \\
Initial Temperature: & 100 K & Equivalence Ratio: & 0.25-3 \\
\end{tabular}
\end{center}

